\documentclass[12pt]{article}

\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}
\newcommand{\answerspace}{\vfill}

\begin{document}

\textbf{Math 126: Calculus II}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 7}\quad \textsl{October 26, 1999}
\hfill\makebox[2.5in][r]{Section:\hrulefill}

\bigskip

1.  Calculate the following, carefully {\bf showing your work} and {\bf
all} important steps.

\medskip
\begin{math}
\sum _{n=1}^\infty (-1)^n e^{-n}.
\end{math}

\answerspace





2. State whether each of the following series converges or diverges.
You {\bf must} give reasons for your answers.

\medskip
a.
\begin{math}
 \sum _{n=2}^\infty \frac {1}{n(\ln(n))^2}.
\end{math}

\answerspace
 
b. 
\begin{math}
  \sum _{n=5}^\infty \frac{n-1}{n+3}.
\end{math}

\answerspace

\end{document}



